Syllabus for Calculus I (Math 141)

Instructor: Michael Filaseta

Office: 301 LeConte

Office Phone Number: 777-6589

Office Hours: By Appointment (afternoons are good)

Text: Calculus with Analytic Geometry (7th ed) by Varberg and Purcell

Chapters to be covered: Chapters 2�6

Grading: quizzes and class participation (10%), 3 tests (20% each), final (30%)

Comments:

(1) The grading scale is as follows. A score of 90% or above will be considered an A, a score of at least 87 % will be considered at least a B+, a score of at least 80 % will be considered at least a B, a score of at least 77 % will be considered at least a C+, a score of at least 70 % will be considered at least a C, a score of at least 67 % will be considered at least a D+, a score of at least 60 % will be considered at least a D, and a score of less than 60 % will usually be considered an F.

(2) The quizzes will be based on the homework. The quizzes will be announced unless it is the impression of the instructor that the homework is not being done by a significant number of the students. If at some point the instructor feels that the homework is not being done properly, he will announce in advance that future quizzes will be given unannounced.

(3) On the quizzes, tests, and the final exam, you are only to use material from Calculus that has been covered in this class prior to that point. For example, L’Hôpital’s Rule for limits is not covered in this course; therefore, you will not be given credit for a problem if you make use of L’Hôpital’s Rule to answer it.

(4) Class participation includes attendence. You are expected to attend every class or to have an appropriate verifiable excuse.

(5) Calculators may be used on homework but are not allowed on quizzes, tests, and the final exam.

(6) Old tests and finals are located in the library on reserve and at:

http://www.math.sc.edu/~filaseta/courses/Math141/Math141.html